Author's response to reviews

Title: Injury in China: A systematic review of injury surveillance studies conducted in Chinese hospital emergency departments

Authors:

Michael Fitzharris (Michael.Fitzharris@monash.edu)
James Yu (jyu@george.org.cn)
Naomi Hammond (nhammond@george.org.au)
Colman Taylor (ctaylor@george.org.au)
Yangfeng Wu (ywu@george.org.cn)
Simon Finfer (sfinfer@george.org.au)
John Myburgh (jmyburgh@george.org.au)

Version: 9 Date: 25 October 2011

Author's response to reviews:

Dear Production team,

Thank-you for formatting the manuscript for publication. I noted 2 things that requiring a fix:
p.29, abbreviations - 'study' was spelt incorrectly (stuy) and have been fixed
p.22, reference 42: a bracket followed the ref number, so it looked like: 42{. ; the bracket has been removed.

Many thanks

Michael